

Citation: Young N, Bowman A, Swedin K, Collins J, Blair-Stahn ND, Lindstedt PA, et al. (2022) Costeffectiveness of antenatal multiple micronutrients and balanced energy protein supplementation compared to iron and folic acid supplementation in India, Pakistan, Mali, and Tanzania: A dynamic microsimulation study. PLoS Med 19(2): e1003902. https://doi.org/10.1371/journal. pmed.1003902

Academic Editor: Jenny E. Myers, University of Manchester, UNITED KINGDOM

Received: March 16, 2021

Accepted: January 4, 2022

Published: February 22, 2022

Copyright: © 2022 Young et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: Data and software used for this study is available at <u>https://github.</u> com/ihmeuw/vivarium_gates_bep.

Funding: The Bill and Melinda Gates Foundation (INV-005218) funded this project. BMGF employees acknowledged in the manuscript assisted in study conceptualization, literature review, methodology, and reviewing the RESEARCH ARTICLE

Cost-effectiveness of antenatal multiple micronutrients and balanced energy protein supplementation compared to iron and folic acid supplementation in India, Pakistan, Mali, and Tanzania: A dynamic microsimulation study

Nicole Young[®]*, Alison Bowman[®], Kjell Swedin[®], James Collins[®], Nathaniel D. Blair-Stahn[®], Paulina A. Lindstedt[®], Christopher Troeger, Abraham D. Flaxman[®]

Institute for Health Metrics and Evaluation, Seattle, Washington, United States of America

* niconico.young@gmail.com

Abstract

Background

Malnutrition among women of childbearing age is especially prevalent in Asia and sub-Saharan Africa and can be harmful to the fetus during pregnancy. In the most recently available Demographic and Health Survey (DHS), approximately 10% to 20% of pregnant women in India, Pakistan, Mali, and Tanzania were undernourished (body mass index [BMI] <18.5 kg/m²), and according to the Global Burden of Disease (GBD) 2017 study, approximately 20% of babies were born with low birth weight (LBW; <2,500 g) in India, Pakistan, and Mali and 8% in Tanzania. Supplementing pregnant women with micro and macronutrients during the antenatal period can improve birth outcomes. Recently, the World Health Organization (WHO) recommended antenatal multiple micronutrient supplementation (MMS) that includes iron and folic acid (IFA) in the context of rigorous research. Additionally, WHO recommends balanced energy protein (BEP) for undernourished populations. However, few studies have compared the cost-effectiveness of different supplementation regimens. We compared the cost-effectiveness of MMS and BEP with IFA to quantify their benefits in 4 countries with considerable prevalence of maternal undernutrition.

Methods and findings

Using nationally representative estimates from the 2017 GBD study, we conducted an individual-based dynamic microsimulation of population cohorts from birth to 2 years of age in India, Pakistan, Mali, and Tanzania. We modeled the effect of maternal nutritional supplementation on infant birth weight, stunting and wasting using effect sizes from Cochrane systematic reviews and published literature. We used a payer's perspective and obtained costs of supplementation per pregnancy from the published literature. We compared disabilitymanuscript. The funders had no role in the decision to publish. ADF was the recipient of this grant.

Competing interests: We have read the journal's policy and the authors of this manuscript have the following competing interests: ADF has consulted recently for Janssen; SwissRe; Sanofi; Merck for Mothers; and Agathos, Ltd. Other authors have no conflicts of interest to disclose.

Abbreviations: ACMR, all-cause mortality rate; ANC, antenatal care; BEP, balanced energy protein; BMI, body mass index; ce, current evidence; CHEERS, Consolidated Health Economic and Evaluation Reporting Standards; CI, confidence interval; CSMR, cause-specific mortality rate; DALY, disability-adjusted life year; DHS, Demographic and Health Survey; EMR, excess mortality rate; GBD, Global Burden of Disease; ICER, incremental cost-effectiveness ratio; IFA, iron and folic acid; IUGR, intrauterine growth restriction; LAZ, length-for-age z-score; LBW, low birth weight; MAL-ED, Etiology, Risk Factors, and Interactions of Enteric Infections and Malnutrition and the Consequences for Child Health; MMS, multiple micronutrient supplementation; PAF, population attributable fraction; RR, relative risk; SGA, small for gestational age; UI, uncertainty interval; WHO, World Health Organization; WLZ, weight-for-length z-score; YLDs, years lived with disability; YLLs, years of life lost.

adjusted life years (DALYs) and incremental cost-effectiveness ratios (ICERs) in a baseline scenario with existing antenatal IFA coverage with scenarios where 90% of antenatal care (ANC) attendees receive either universal MMS, universal BEP, or MMS + targeted BEP (women with prepregnancy BMI <18.5 kg/m² receive BEP containing MMS while women with BMI >18.5 kg/m² receive MMS). We obtained 95% uncertainty intervals (UIs) for all outputs to represent parameter and stochastic uncertainty across 100 iterations of model runs. ICERs for all scenarios were lowest in Pakistan and greatest in Tanzania, in line with the baseline trend in prevalence of and attributable burden to LBW. MMS + targeted BEP averts more DALYs than universal MMS alone while remaining cost-effective. ICERs for universal MMS compared to baseline IFA were \$52 (95% UI: \$28 to \$78) for Pakistan, \$72 (95% UI: \$37 to \$118) for Mali, \$70 (95% UI: \$43 to \$104) for India, and \$253 (95% UI: \$112 to \$481) for Tanzania. ICERs for MMS + targeted BEP compared to baseline IFA were \$54 (95% UI: \$32 to \$77) for Pakistan, \$73 (95% UI: \$40 to \$104) for Mali, \$83 (95% UI: \$58 to \$111) for India, and \$245 (95% UI: \$127 to \$405) for Tanzania. Study limitations include generalizing experimental findings from the literature to our populations of interest and using population-level input parameters that may not reflect the heterogeneity of subpopulations. Additionally, our microsimulation fuses multiple sources of data and may be limited by data quality and availability.

Conclusions

In this study, we observed that MMS + targeted BEP averts more DALYs and remains costeffective compared to universal MMS. As countries consider using MMS in alignment with recent WHO guidelines, offering targeted BEP is a cost-effective strategy that can be considered concurrently to maximize benefits and synergize program implementation.

Author summary

Why was this study done?

- Iron and folic acid (IFA) supplementation and IFA-containing multiple micronutrient supplementation (MMS) taken during pregnancy can improve the health of newborns, particularly by increasing their birth weight, which reduces the risk of infant mortality.
- For pregnant mothers who are undernourished, increased caloric intake through balanced energy protein (BEP) supplementation can provide an additional benefit on birth weight and its associated health outcomes.
- We conducted this study to compare the costs and benefits of BEP, MMS, and IFA supplementation in areas burdened by low birth weight (LBW) and its associated health detriments.

What did the researchers do and find?

- We estimated the impact of IFA, MMS, and BEP supplementation in India, Pakistan, Mali, and Tanzania by developing an individual-based dynamic microsimulation model of populations of newborns that tracked their health outcomes during the first 2 years of life.
- We found that supplying MMS to 90% of women who attend antenatal care (ANC) visits resulted in improving morbidity and mortality among children under 2 at a cost of \$52 per disability-adjusted life year (DALY) saved for Pakistan, \$72 for Mali, \$70 for India, and \$253 for Tanzania.
- Supplying BEP (containing MMS) to 90% of women who attend ANC with low body mass index (BMI) and MMS to 90% of women who attend ANC with adequate BMI saved more DALYs than supplying 90% of all women who attend ANC visits with MMS and was similarly cost-effective at \$54 per DALY saved relative to existing IFA practice for Pakistan, \$73 for Mali, \$83 for India, and \$245 for Tanzania.

What do these findings mean?

- High coverage of MMS for women at ANC is a cost-effective strategy to improve child health outcomes relative to the existing coverage and practice of supplying only IFA.
- While supplying BEP (containing MMS) universally to women at ANC has the greatest impact on child health outcomes, targeting BEP for undernourished women and offering MMS to women of adequate BMI is more cost-effective and has greater impact on child health outcomes than supplying MMS alone.
- Our findings are limited by applying effects of maternal supplementation regimens from clinical trials to the modeled populations, and we did not model subnational variation.

Introduction

Pregnancy requires increased energy and nutritional consumption to support the body's changing physiology and fetal growth [1]. Without proper nourishment during pregnancy, the fetus may fail to reach adequate growth, resulting in intrauterine growth restriction (IUGR) [2]. IUGR can lead to babies born small for gestational age (SGA), defined as weight below the 10th percentile for gestational age or with low birth weight (LBW), defined as birth weight below 2,500 g. IUGR and LBW are associated with increased risk of neonatal death, infection, neurological impairment, wasting, and stunting in childhood as well as increased risk of chronic disease, diminished scholastic achievement, lower income, and decreased birth weight of offspring in adulthood [3,4]. Asia and Africa contain the greatest burden of LBW: In 2015, an estimated 48% and 24% of all affected newborns globally were from Asia and sub-Saharan Africa, respectively [5]. These regions also have the highest levels of micronutrient deficiencies among pregnant women and women of reproductive age [6]. Ensuring adequate maternal

nutrition during gestation is a necessary and worthwhile investment that will have far-reaching benefits for current and future generations [7].

Supplementing women with micro and macronutrients during pregnancy can improve birth outcomes. Recently, the World Health Organization (WHO) updated its 2016 recommendation to use multiple micronutrient supplementation (MMS) that include IFA and additional 13 to 15 micronutrients in the context of rigorous research [8,9]. Cost-effectiveness analysis in some countries has shown that replacing IFA with MMS is highly cost-effective at under \$100 per disability-adjusted life year (DALY) averted [10-13]. Currently, WHO recommends balanced energy protein (BEP) food supplements in settings with high prevalence of undernourished pregnant women (over 20%) [14]. BEP food supplements are foods in which protein provides less than 25% of the total energy content. They can come in several forms such as biscuits, beverages, or sachets and can be made using locally sourced ingredients [15]. The latest Cochrane review from 2015 found that use of BEP supplements during pregnancy can reduce the risk of stillbirth (relative risk [RR] = 0.6, 95% UI 0.39 to 0.94) and SGA (RR = 0.7, 95% UI 0.69 to 0.9) as well as increase mean birth weight (+40.96 g, 95% UI: 4.66 to 77.26) [15]. However, BEP's potential may have been under realized in the trials due to heterogeneity in the BEP supplement compositions, supplementation period, and comparison groups used. Moreover, nonadherence or dietary substitution might have led to only small net increases in energy intake in some of the trials [15]. Greater effect sizes for BEP on birth weight have been reported by some studies, potentially due to study populations with larger energy deficits or use of a higher energy supplement [16-18]. Additionally, BEP may improve wasting and stunting in the first 5 years of life [19], but more studies are needed to draw conclusions on BEP's long-term impacts on child growth [20]. Better optimized formulations of BEP that consist of equivalent micronutrients to MMS are available and may address these gaps [21]. These formulations create an integrated delivery vehicle that can improve acceptability and adherence and are currently under clinical trials.

One challenge to BEP implementation is its higher cost compared to IFA and MMS. One study estimated that BEP costs more than \$500 per DALY averted, which may be over the affordability thresholds for some low- and middle-income countries [22]. A potentially more cost-effective strategy is to target BEP to undernourished pregnant women. Subgroup analysis comparing undernourished and adequately nourished women suggests a greater increase in birth weight among undernourished women (+66.96 g, 95% UI: 13.13 to 120.78) than adequately nourished women (+15.93 g, 95% UI: -20.83 to 52.69) [15]. Moreover, prevalence of obesity and overweight is increasing, and women with overweight or obesity are at higher risk of fetal overgrowth and macrosomia (\geq 4,000 g) [23]. Risks of supplementing high body mass index (BMI) women with BEP are still unknown [14]. Targeting BEP to undernourished women for BEP does not fall under the 2016 WHO antenatal recommendation, we explored this approach in our simulation to understand its potential impact.

Maternal MMS and BEP supplementation are 2 of 10 critical direct nutritional interventions identified by the 2013 *Lancet* Series on maternal and child nutrition [24,25], but to the best of our knowledge, there is a lack of cost-effectiveness analysis comparing these interventions [26]. Leveraging estimates from the Global Burden of Disease (GBD) 2017 study, we used dynamic microsimulation to compare the impact on DALYs and the cost-effectiveness of supplementing pregnant women with (1) universal MMS; (2) universal BEP; or (3) MMS + targeted BEP (where undernourished women receive BEP supplements containing MMS with IFA and adequately nourished women receive MMS containing IFA) with current baseline coverage of IFA in India, Pakistan, Tanzania, and Mali. Although these countries have less than 20% prevalence of undernourished pregnant women, WHO criteria for universal BEP, they have high burdens of LBW, an outcome that BEP improves. Our model predicts the impact of several scenarios to inform government or donor decision-making on MMS and BEP supplementation for pregnant women.

Methods

Conceptual framework and model logic

We developed an individual-based dynamic microsimulation model in Python 3.6 using the Vivarium simulation framework [27] to compare the DALYs averted and costs of various maternal supplementation scenarios among the population under 2 years old in the 4 countries. We chose a dynamic mathematical model to capture the complex effects of maternal BMI and different supplementation scenarios on child development in the first 2 years of life. The model components included in the simulation and the relationships between them were planned a priori as informed by evidence from the GBD study and other literature sources. We modeled the effect of maternal BMI on birth weight and the effect of supplementation on birth weight, stunting, and wasting. We did not include the effect of supplementation on maternal risk factors. In our model, the effect of LBW and short gestation acts directly on mortality in the neonatal period. The effect of stunting and wasting acts on mortality and morbidity through protein energy malnutrition (for wasting only), measles, diarrheal diseases, and lower respiratory tract infections. These risk-outcome relationships were selected according to the GBD 2017 study risk-outcome pair criteria for convincing or probable evidence of causation [28] (Fig 1). Model implementation strategy was adapted and iteratively updated when model verification criteria were not met throughout model development. The iterative workflow process followed throughout model development is described in S1 Supplement. Parameter values for the intervention costs, the association between maternal BMI and birth weight, the hypothesized effect size of BEP for undernourished mothers in a best-case scenario, and the sensitivity analysis were revised after peer review, and the model was rerun.

For each country, we simulated a population of 100,000 live newborns with a model run length of 2 years. In accordance with the 2017 GBD study, birth weight exposure affects health outcomes during the neonatal period (0 to 27 days), and wasting and stunting exposures affect health outcomes from the neonatal period until 5 years of life. We chose a 2-year time horizon to capture the effect of in utero maternal supplementation exposure on child health outcomes in early life. We used fixed time increments of 1 day as the time step. We obtained 95% uncertainty intervals (UIs) to represent parameter and stochastic uncertainty across 100 iterations of model runs.

At model initialization, we assigned attributes to simulants including age (0 at birth), sex, maternal undernourishment status, supplementation coverage, risk exposures (birth weight as well as wasting and stunting z-scores), and possession of any congenital disease or conditions. We used location-, age-, and sex-specific GBD 2017 national average estimates as inputs for the population sex structure, risk exposure, and disease and mortality model components [28–31]. Details of maternal undernourishment status, coverage, and risk exposures are described below in the baseline model section. The attributes of each individual at the start of the simulation determined disease states, mortality, and morbidity at subsequent time steps using Monte Carlo sampling from model parameters. RRs, the population attributable fraction (PAF) between risk–outcome pairs, and population incidence rates determined probability of disease or condition (for detailed methodology of RR and PAF calculations, see Supplementary Appendix 1 of GBD 2017 Risk Factor Study [28]). Briefly, the RRs for a risk–outcome pair represent the risk of disease for a given exposure level relative to the theoretical minimum risk exposure level (TMREL;RR = 1 for TMREL).



Fig 1. Model concept diagram. Logical causal flow of maternal characteristics and antenatal nutritional supplementation regimens on downstream effects and ultimate impact on mortality and morbidity in our microsimulation model. In GBD 2017, LBW is modeled to directly affect all-cause mortality among early (0 to 6 days) and late neonatal (7 to 27 days) age groups. Gray arrows denote effects with potential but inconclusive evidence. GBD, Global Burden of Disease; LAZ, length-for-age z-score; LBW, low birth weight; WLZ, weight-for-length z-score.

https://doi.org/10.1371/journal.pmed.1003902.g001

Incidence rate among exposure category = Population incidence*(1-PAF)*RR Incidence rate among TMREL = Population incidence*(1-PAF) Probability of disease or condition within time step = $1-e^{(-incidence\ rate*\ time_step)}$

A simulant's cause state (x_i , where $x_1 = 1$ represents with disease or condition, and $x_0 = 0$ represents without disease or condition), the all-cause mortality rate (ACMR), as well as cause-specific mortality rates (CSMRs), and excess mortality rates (EMRs) for modeled disease or conditions were used to determine the simulant's mortality probability. The time of a simulant's death in relation to the theoretical maximum life expectancy at that age determines years of life lost (YLLs).

Mortality hazard = ACMR + $\sum_{cause=x}^{n} (-CSMR + EMR(x_i))$, where x = disease or conditions in Fig 1.

Probability of mortality within time step = $1 - e^{(-mortality hazard^* time_step)}$

Alive simulants may recover from a disease or condition according to the cause-specific remission rate. Disability weights associated with the disease or condition determined simulants' years lived with disability (YLDs) [30]. We calculated DALYs by summing YLLs and YLDs.

We compared DALYs from different intervention scenarios with the DALYs from the scenario with baseline coverage of IFA supplementation. For intervention scenarios, we scaled up supplementation coverage to 90% where routine antenatal care (ANC) attendees receive (1) universal MMS; (2) universal BEP; or (3) MMS + targeted BEP (attendees with low prepregnancy BMI (<18.5 kg/m²) receive BEP containing MMS, while those above (BMI >18.5 kg/m²) receive MMS). We applied common random numbers between scenarios to ensure that differences in outcomes were due to the different interventions rather than stochastic variation. Data and software are held at https://github.com/ihmeuw/vivarium_gates_bep.

Baseline model

Maternal undernourishment. Maternal undernourishment is a risk factor for LBW [7]. We stratified children into those born to (1) undernourished mothers (BMI <18.5 kg/m²); and (2) adequately nourished mothers (BMI \ge 18.5 kg/m²). We obtained the prevalence of maternal undernourishment from the Demographic and Health Survey (DHS) and ongoing BEP trials where DHS data were unavailable (Table 1).

We assumed a crude birth weight difference of 138.46 g with a 95% confidence interval (CI) of 174.68 to 102.25 between maternal BMI strata, with higher birth weight babies born to mothers with BMI >18.5 kg/m². We obtained this effect difference by conducting a literature search to identify birth weight effect sizes and pooling across 18 studies using a random effects model (S2 Supplement).

IFA baseline coverage. We obtained baseline IFA coverage from the latest available DHS (Table 1). We stratified the population into those who received IFA and those who did not. We assumed that infants born to IFA supplemented mothers had an average birth weight of +57.73g (95% CI: 7.66 to 107.79) greater than infants born to mothers not supplemented with IFA [32]. We assumed all women who were supplemented during pregnancy received supplements through ANC visits. We used the coverage proportion for pregnant women who received ANC from a skilled provider at least one time from GBD 2017 as the proportion of women who attend ANC visits and are thus able to receive supplementation.

We therefore have 4 subgroups of birth weights broken out from the GBD 2017 population mean birth weight such that the 4 groups' (2 IFA strata by 2 BMI strata) weighted average equals the population mean. The difference in birth weight between groups reflects the mean effect differences we obtained from the literature.

Birth weight and child growth. In GBD 2017, the birth weight and gestational age exposure was modeled as an ordered polytomous joint distribution specifying the prevalence of births in 500 g by 2-week birth weight–gestational age categories [28]. We first converted this discrete exposure distribution into a continuous joint exposure distribution of birth weight and gestational age by assuming a uniform distribution of birth weights and gestational ages

Country	ANC from skilled provider [†]	Women who took antenatal iron for 90+ days [‡]	Women who are undernourished according to BMI (<18.5 kg/m ²)		
	(%)	(%)	(%)		
India	88.2 (87.4 to 89.0)	38.7 (31.0 to 46.4)	16.8 (13.4 to 20.2) ⁹		
Pakistan	84.4 (80.5 to 87.8)	29.4 (23.5 to 35.3)	10.7 (8.6 to 12.8) ⁹		
Mali	83.5 (79.1 to 87.5)	28.0 (22.4 to 33.6)	10.3 (8.2 to 12.4) ^µ		
Tanzania	98.3 (97.7 to 98.8)	21.4 (17.1 to 25.7)	9.5 (7.6 to 11.4) ^µ		

Table 1. Baseline coverage of ANC from skilled provider, coverage of IFA, and proportion undernourished by country with 95% CIs.

Note: Where 95% confidence ranges were unavailable, we applied a range of $\pm 20\%$ from the reported mean value as a plausible coefficient of variability.

 $^{\dagger}\mathrm{GBD}$ 2017: Proportion of pregnant women receiving any ANC from a skilled provider.

^{*}Most recent available DHS: percentage of women with a live birth in the 5 (or 3) years preceding the survey who took 90+ days of iron tablets or syrup during ANC. ^µMost recent available DHS, among women aged 15 to 49.

⁹Ongoing Gates trials, among pregnant women.

ANC, antenatal care; BMI, body mass index; CI, confidence interval; DHS, Demographic and Health Survey; GBD, Global Burden of Disease; IFA, iron and folic acid.

https://doi.org/10.1371/journal.pmed.1003902.t001

within each category. We assigned each simulant a continuously distributed birth weight and gestational age, which was mapped back to the appropriate risk category in GBD to obtain the RRs for all-cause mortality for the early neonatal (0 to 6 days) and late neonatal (7 to 27 days) age groups. The RRs used in this analysis were specific to each 500 g by 2-week birth weight–gestational age category such that the difference in mortality risk between 2 categories with different birth weights and the same gestational age varied across gestational age categories. Additionally, the difference in mortality risk from one birth weight category to the next with constant gestational age varied such that the magnitude of mortality risk reduction tended to be greater among the lower end of the birth weight distribution.

We used a propensity model for wasting and stunting. Each simulant was initialized with a "propensity" for wasting and stunting, and the simulant's z-scores were determined by comparing this propensity to the overall age-specific z-score exposure prevalence in the population such that individual simulant's wasting and stunting z-scores may change as they age, but their population percentile remains constant. The TMREL for wasting and stunting were defined as z-scores greater than -1 SD of WHO 2006 standard weight-for-length and length-for-age curves, respectively.

We used a correlation Spearman's rho of 0.394 (95% UI: 0.353 to 0.433; SD: 0.020) for birth weight and length-for-age z-score (LAZ) at 6 months and Spearman's rho of 0.308 (95% CI: 0.263 to 0.351; SD: 0.022) for birth weight and weight-for-length z-score (WLZ) at 6 months. The correlation coefficients were obtained from a pooled analysis of multicountry birth cohorts from the Etiology, Risk Factors, and Interactions of Enteric Infections and Malnutrition and the Consequences for Child Health (MAL-ED) study [33].

Intervention model

We modeled 3 intervention scenarios where 90% of ANC attendees received (1) universal MMS; (2) universal BEP; or (3) MMS + targeted BEP (Fig 2). We assume the formulation of MMS contains IFA and that BEP contains MMS, which contains IFA. In each scenario, IFA,



Fig 2. Visual diagram of supplementation coverage among women who attend ANC. Baseline IFA: scenario with countryspecific reported coverage of IFA consumption; Universal MMS: scenario where 90% of antenatal attendees receive MMS; Universal BEP: scenario where 90% of antenatal attendees receive BEP; MMS + targeted BEP: scenario where 90% of antenatal attendees receive BEP if undernourished and receive MMS if adequately nourished according to prepregnancy BMI. *We modeled BEP scenarios twice: once with effects supported by ce and second with he suggested by some evidence. ANC, antenatal care; BEP, balanced energy protein (contains MMS); BMI, body mass index; ce, current evidence; he, hypothesized evidence; IFA, iron and folic acid; MMS, multiple micronutrient supplementation (contains IFA).

https://doi.org/10.1371/journal.pmed.1003902.g002

Supplementation	Infant outcomes	Comparison groups	Mean difference (95% CI)	Cost per beneficiary in 2021 USD (95% $\text{CI})^{\dagger}$		
IFA	Birth weight (g)	Unsupplemented	+57.73 g (7.66 to 107.79) (32)	17.37 (13.90 to 20.84) (13)		
Multiple micronutrient	ient Birth weight (g) Iron with or without folic +45.16 g (32.31 to 58.02) (9) acid		19.47 (15.58 to 23.36) (13)			
BEP (ce)	Birth weight (g)	Control or no intervention*	<i>Undernourished</i> : +66.96 g (13.13 to 120.78) (15)	India: 47.88 (38.30 to 57.46) (34) Pakistan: 45.96 (36.77 to 55.15) (34)		
			<i>Adequately nourished</i> : +15.93 g (-20.83 to 52.69) (15)	Mali: 41.20 (32.96 to 49.44) (34) Tanzania: 41.90 (33.52 to 50.28) (34)		
BEP (he^{π})	Birth weight (g)	Control or no intervention*	<i>Undernourished</i> : +136 g (79 to 193)** (16)			
			<i>Adequately nourished</i> : +15.93 g (-20.83 to 52.69) (15)			
$\begin{array}{c} \text{EP (he}^{\pi}) \\ \text{BEP (he}^{\pi}) \\ \text{months} \end{array} \qquad $		+0.3 (±0.1) (19)				
BEP (he ^π)	WLZ score at 6	Unsupplemented	+0.3 (±0.1) (19)			

Table 2. Effects of different maternal intervention on birth weight and 6-month child growth indicators.

Note: In each scenario, we applied effect sizes additively according to nutrients received.

*The trials from the 2015 Cochrane review compared a range of comparison groups. We interpreted the effect size as that of balanced energy plus vitamins and minerals versus vitamins and minerals.

**CI calculated from reported standard error as $\pm 1.96 \times SE$.

 π We modeled 2 additional scenarios for BEP: once with effects supported by ce and second with he suggested by some evidence.

BEP, balanced energy protein; ce, current evidence; CI, confidence interval; he, hypothesized evidence; IFA, iron and folic acid; LAZ, length-for-age z-score; WLZ, weight-for-length z-score.

https://doi.org/10.1371/journal.pmed.1003902.t002

MMS, and BEP supplementations were modeled to improve infant birth weight as supported by current evidence (ce) [9,15,32]. For BEP, we modeled a differential effect of BEP on birth weight by maternal nourishment status because there is some evidence to suggest that BEP may have a greater impact on birth weight among women with low prepregnancy body weight [15].

For interventions involving BEP, we modeled 2 additional scenarios with BEP hypothesized effects (he) on WLZ and LAZ scores at 6 months as suggested in Kusin and colleagues [19] and greater effect sizes for BEP on birth weight for undernourished women as suggested by some studies as a best-case scenario [16–18] (Table 2).

Sensitivity analysis

We performed a sensitivity analysis around supplementation coverage in which we modeled the impact of each intervention scenario with coverage levels equal to the location-specific proportion of women who received IFA supplementation at baseline (Table 1).

Data sources

Inputs parameterized to the GBD study. We used 1,000 draw-level location/age/sex-specific risk exposure distributions, disease prevalence at birth, disease incidence rate, CSMR, ACMR, and EMR from the GBD 2017 study to parameterize the population-level rates in our model. We assigned individual simulant-level measures in the baseline scenario according to coverage of IFA supplementation and maternal underweight such that (1) the birth weight and downstream outcome differences (**Fig 1**) among IFA-supplemented and unsupplemented individuals as well as undernourished and adequately nourished mothers reflected the effect sizes of IFA and maternal undernourishment; and (2) the population-level measures in our model reflected the GBD 2017 estimates.

Effect sizes and costs from published literature. Table 2 presents the interventions, affected outcomes, effect sizes, and cost per supplementation course. We assumed a normal distribution of uncertainty for effect sizes. For the effect of MMS on birth weight, we reviewed studies from Keats and colleagues' Cochrane review to obtain birth weight mean differences between MMS supplemented groups and iron with or without folic acid supplemented groups [9]. A total of 13 out of 18 trials reported mean birth weight shifts. We pooled the birth weight differences using a random effects model (S3 Supplement). For each scenario, we applied effect sizes additively according to supplements received.

For our cost analysis, we obtained the IFA and MMS costs from Kashi and colleagues [13]. These estimates considered costs of supplements, service delivery, and program expenses. We obtained costs for BEP supplementation from Scott and colleagues [34]. The authors calculated country-specific BEP costs by scaling the range of costs from Shekar and colleagues [35]. All costs reflect 180 days (6 months) of supplementation during pregnancy. The costs reflect a payer's perspective as results from this study are intended to aid government or donor decision-making. Where there was an absence of quantified UIs from available sources, we assumed a plausible coefficient of variation of 20% about the point estimates. We did not include discounting as the ce on the benefits of supplementation occur mostly in the neonatal period, within 1 year of the supplementation period. We use half of GDP per capita as the cost-effective threshold for acceptable cost per DALY averted [22]. All costs were adjusted to August 2021 USD using the United States Bureau of Labor Statistics Consumer Price Index Inflation Calculator (accessed September 2021) [36].

DALYs and ICERs

We summed YLLS and YLDs to obtain DALYs per 100,000 births for each scenario. We then compared each scenario's DALYs to the reference baseline scenario's DALYs to calculate averted DALYs. We tracked counts of antenatal supplement courses administered to calculate the total cost of the intervention for each scenario. We calculated the incremental cost-effectiveness ratios (ICERs) by dividing the difference in costs between 2 interventions (cost difference) by the difference in their benefits (DALYs averted).

Model verification

We verified the results of our model, in particular the ACMRs, CSMRs, disease state transition rates, and risk exposure distributions against the estimates from GBD 2017 to confirm that the model estimates matched. We also verified the model's maternal undernourishment and supplementation effect sizes with the input data.

This study is reported as per the Consolidated Health Economic and Evaluation Reporting Standards (CHEERS) Statement (<u>S4 Supplement</u>).

Results

Model verification

We compared the output parameters from our baseline model to the GBD 2017 input estimates (<u>S5 Supplement</u>). All model outputs were within 10% of GBD estimates used as model inputs, namely ACMRs, CSMRs, cause incidence, prevalence, and remission. Effect sizes between maternal nourishment strata and supplementation strata also matched our input data.

	India	Pakistan	Mali	Tanzania
Birth weight				
Mean birth weight, grams				
Overall	2,889 (2,880 to 2,899)	2,843 (2,818 to 2,876)	2,930 (2,899 to 2,993)	3,268 (3,190 to 3,320)
Maternal undernourished	2,771 (2,742 to 2,805)	2,718 (2,666 to 2,763)	2,803 (2,762 to 2,869)	3,140 (3,062 to 3,211)
Maternal adequately nourished	2,912 (2,903 to 2,926)	2,858 (2,834 to 2,891)	2,944 (2,909 to 3,010)	3,282 (3,200 to 3,336)
Prevalence of LBW	21.5% (20.8% to 22.1%)	23.6% (21.6% to 25.5%)	18.0% (15.5% to 19.7%)	7.7% (6.3% to 9.9%)
Wasting				
Mean WLZ at 6 m	-0.428 (-0.456 to -0.401)	-0.460 (-0.604 to -0.347)	-0.408 (-0.541 to -0.301)	-0.090 (-0.156 to -0.044)
% Not wasted	63.8% (63.4% to 64.2%)	67.8% (63%.7 to 71.0%)	67.9% (64.8% to 70.6%)	77.4% (75.5% to 78.8%)
% Mildly wasted (WLZ < -1)	19.6% (19.3% to 20.0%)	21.8% (19.4% to 25.1%)	20.3% (18.3% to 23.7%)	15.8% (15.1% to 16.7%)
% Moderately wasted (WLZ < -2)	10.6% (10.3% to 10.8%)	7.9% (6.2% to 9.5%)	8.4% (7.4% to 9.7%)	5.2% (4.6% to 5.9%)
% Severely wasted (WLZ < -3)	6.0% (5.8% to 6.3%)	2.6% (1.3% to 4.5%)	3.3% (2.1% to 4.7%)	1.6% (1.1% to 2.0%)
Stunting				
Mean LAZ at 6 m	-0.887 (-0.935 to -0.829)	-1.388 (-2.015 to -0.819)	-0.853 (-1.370 to -0.446)	-1.081 (-1.495 to -0.777)
% Not stunted	51.0% (50.1% to 52.1%)	36.4% (16.1% to 50.6%)	52.8% (30.2% to 66.0%)	45.6% (27.6% to 54.5%)
% Mildly stunted (LAZ < -1)	20.4% (19.6% to 21.4%)	26.1% (15.7% to 45.8%)	34.2% (21.2% to 55.9%)	30.2% (21.7% to 48.2%)
% Moderately stunted (LAZ $<$ –2)	15.0% (14.6% to 15.5%)	20.7% (13.6% to 32.9%)	10.6% (7.6% to 25.5%)	17.0% (14.0% to 24.0%)
% Severely stunted (LAZ < -3)	13.6% (12.9% to 14.4%)	16.8% (8.0% to 30.4%)	2.4% (0.5% to 4.7%)	7.8% (4.1% to 10.7%)

Table 3. Characteristics of baseline simulated populations of the 4 modeled countries with 95% CIs.

Note: RRs for GBD risk–outcome pairs are above 1 for z-scores -1 to -3.

CI, confidence interval; GBD, Global Burden of Disease; LAZ, length-for-age z-score; LBW, low birth weight defined as <2,500 g; m, months; RR, relative risk; WLZ, weight-for-length z-score.

https://doi.org/10.1371/journal.pmed.1003902.t003

Characteristics of baseline population

Table 3 presents the mean birth weight, prevalence of LBW, prevalence of wasting and stunting, and their mean z-scores in the baseline population of our simulation. **Table 4** presents the baseline percentage of DALYs attributable to LBW, stunting, and wasting in our simulation. Tanzania had substantially higher mean birth weight and mean WLZ-scores than the other countries at baseline.

DALYs, ICERs, and cost-effectiveness thresholds

Our model yielded baseline burdens of 324,000 (95% UI: 307,000 to 343,000), 456,000 (95% UI: 376,000 to 543,000), 723,000 (95% UI: 600,000 to 872,000), and 406,000 (95% UI: 322,000 to 486,000) DALYs per 100,000 live births for India, Pakistan, Mali, and Tanzania, respectively (**Fig 3**). Assuming the proportion of women attending ANC remains unchanged and 90% of ANC attendees receive supplementation in our intervention scenarios, universal BEP averts the most DALYs followed by MMS + targeted BEP and then universal MMS compared to baseline IFA (**Fig 4**). Scenarios using the hypothesized effects for BEP resulted in more DALYs averted relative to baseline than the scenarios using the ce (**Fig 4**). The intervention effects per 100,000 live births are strongest in Pakistan, followed by Mali, India, and Tanzania. This follows the pattern in LBW prevalence and percentage of baseline DALYs attributable to LBW (**Table 4**).

The ICERs for universal MMS compared to baseline IFA were lowest for Pakistan, followed by India, Mali, and Tanzania at \$52 (95% UI: 28, 78), \$70 (95% UI: 43, 104), \$72 (95% UI: 37, 118), and \$253 (95% UI: 112, 481), respectively. The ICERs for MMS + targeted BEP relative to baseline IFA using ce effects for BEP were similar to those of universal MMS for each

	India			Pakistan		Mali			Tanzania			
Cause attributable to	% of DALYs	Risk attribution	DALYs	% of DALYs	Risk attribution	DALYs	% of DALYs	Risk attribution	DALYs	% of DALYs	Risk attribution	DALYs
LBW												
All causes* affected	20.42%	49.27%	10.06%	39.01%	54.16%	21.13%	37.48%	45.17%	16.93%	30.38%	40.88%	12.42%
Total DALYs attributable			10.06%			21.13%			16.93%			12.42%
Stunting												
Measles	0.30%	21.23%	0.06%	0.17%	19.54%	0.03%	0.95%	15.79%	0.15%	0.69%	20.72%	0.14%
Diarrhea	4.71%	3.80%	0.18%	5.47%	9.72%	0.53%	6.30%	6.74%	0.42%	3.98%	5.35%	0.21%
Lower respiratory tract infection	4.26%	7.06%	0.30%	5.69%	11.33%	0.64%	8.12%	10.58%	0.86%	9.06%	11.67%	1.06%
Total DALYs attributable			0.54%			1.21%			1.43%			1.41%
Wasting												
Measles	0.30%	41.92%	0.13%	0.17%	33.86%	0.06%	0.95%	35.26%	0.33%	0.69%	22.13%	0.15%
Diarrhea	4.71%	21.33%	1.00%	5.47%	51.12%	2.80%	6.30%	52.22%	3.29%	3.98%	26.24%	1.04%
Lower respiratory tract infection	4.26%	31.31%	1.33%	5.69%	46.87%	2.67%	8.12%	62.04%	5.04%	9.06%	43.51%	3.94%
Protein energy malnutrition	0.57%	100.00%	0.57%	1.00%	100.00%	1.00%	9.00%	100.00%	9.00%	2.23%	100.00%	2.23%
Total DALYs attributable			3.03%			6.52%			17.66%			7.37%

Table 4. Baseline DALYs attributable to modifiable risk factors among male and female children of under 2 years of age, 2017.

*Causes of morbidity and mortality affected by LBW according to the 2017 GBD study criteria for causality include neonatal preterm birth, neonatal encephalopathy, lower respiratory tract infection, other neonatal disorders, neonatal sepsis, hemolytic disease and jaundice, upper respiratory tract infections, otis media, diarrhea, meningitis, and encephalitis.

DALY, disability-adjusted life year; GBD, Global Burden of Disease; LBW, low birth weight.

https://doi.org/10.1371/journal.pmed.1003902.t004



Fig 3. DALYs due to all causes among children under 2 in thousands per 100,000 live births in each simulated supplementation scenario by location. The width of the colored boxes represents the interquartile range (25th to 75th percentiles); the solid line in the box represents the median (50th percentile); the whiskers on either end of the box represent the minimum and maximum values; and the points represent outlier values. BEP, balanced energy protein; ce, scenarios with current evidence effects for BEP; DALY, disability-adjusted life year; he, scenarios with hypothesized effects for BEP; IFA, iron and folic acid; MMS, multiple micronutrient supplementation.

https://doi.org/10.1371/journal.pmed.1003902.g003



Fig 4. DALYs averted in thousands relative to the baseline scenario for each simulated supplementation intervention scenario among children under 2 per 100,000 live births by location. The width of the colored boxes represents the interquartile range (25th to 75th percentiles); the solid line in the box represents the median (50th percentile); the whiskers on either end of the box represent the minimum and maximum values; and the points represent outlier values. BEP, balanced energy protein; ce, scenarios with current evidence effects for BEP; DALY, disability-adjusted life year; he, scenarios with hypothesized effects for BEP; MMS, multiple micronutrient supplementation.

https://doi.org/10.1371/journal.pmed.1003902.g004

respective location at \$54 (95UI: 32, 77), \$83 (95% UI: 58, 111), \$73 (95% UI: 40, 104), and \$245 (95% UI: 127, 405). Universal BEP using ce for BEP effects was less cost-effective than universal MMS and MMS + targeted BEP scenarios (Table 5, Fig 5). For both the MMS + targeted BEP and universal BEP, ICERs using the hypothesized effects of BEP were lower than those using the ce for the effects of BEP. For Mali and Tanzania, universal BEP relative to baseline IFA using hypothesized effects for BEP had lower ICERs than universal MMS at \$62 (95% UI: 41, 94) and \$133 (95% UI: 88, 195), respectively. For each modeled location, universal MMS and MMS + targeted BEP ICERS were less than half of GDP (Table 6).

Offering BEP to all ANC attendees was least cost-effective, costing between \$121 and \$487 per DALY (mean estimates) for the modeled countries based on ce effect sizes (**Figs 5 and 6**). **Table 6** presents the DALYs averted, numbers treated, total cost of treatment, and ICERs per 100,000 live births for each scenario.

Sensitivity analysis of supplementation coverage

When intervention coverage was implemented at the same level of baseline IFA coverage, fewer DALYs were averted per 100,000 births, and ICERs were lower in comparison to intervention coverage of 90% of ANC attendees. In this sensitivity analysis, the universal MMS scenario resulted in 3,746 (95% UI: 2,238, 5,846), 4,144 (95% UI: 2,284, 6,904), 2,950 (95% UI: 1,492, 4,796), and 662 (95% UI: 176, 1,364) DALYs averted relative to the baseline scenario per 100,000 live births for India, Pakistan, Mali, and Tanzania, respectively. Likewise, the MMS + targeted BEP scenario using ce effect sizes resulted in 4,974 (95% UI: 3,158, 7,033), 5,168 (95% UI: 2,854, 7,956), 3,685 (95% UI: 1,977, 5,674), and 864 (95% UI: 217, 1,669) DALYs averted for the respective locations.

The universal MMS intervention was more cost-effective in the sensitivity analysis than coverage for 90% of ANC attendees at \$23 (95% UI: 13, 39), \$16 (95% UI: 9, 26), \$22 (95% UI: 12, 37), and \$72 (95% UI: 34, 292) per DALY averted relative to baseline for India, Pakistan, Mali, and Tanzania, respectively. Likewise, the BEP interventions were slightly more cost-

	DALYs averted relative to baseline per 100,000 live births	Number treated per 100,000 live births	Total cost (2021 USD) per 100,000 live births	ICER (2021 USD per DALY averted) relative to baseline
India				
Baseline IFA	-	38,829 (25,830 to 42,930)	676,981 (536,418 to 830,096)	-
Universal MMS	13,014 (9,220 to 18,735)	79,402 (78,593 to 80,294)	1,551,238 (1,200,889 to 1,855,065)	70 (43 to 104)
Universal BEP (ce)	17,396 (11,035 to 23,875)	79,402 (78,593 to 80,294)	3,814,756 (2,953,189 to 4,561,916)	188 (120 to 285)
MMS + targeted BEP (ce)	15,584 (11,767 to 20,883)	MMS: 65,960 (64,540 to 67,603), BEP: 13,442 (12,247 to 14,758)	1,934,423 (1,487,349 to 2,322,707)	83 (58 to 111)
Universal BEP (he)	30,504 (21,526 to 38,574)	79,402 (78,593 to 80,294)	3,814,756 (2,953,189 to 4,561,916)	105 (72 to 151)
MMS + targeted BEP (he)	19,186 (15,064 to 24,467)	MMS: 65,960 (64,540 to 67,603), BEP: 13,442 (12,247 to 14,758)	1,934,423 (1,487,349 to 2,322,707)	67 (47 to 88)
Pakistan				
Baseline IFA	-	29,484 (27,216 to 32,572)	514,079 (406,470 to 631,709)	-
Universal MMS	19,931 (12,385 to 32,270)	75,852 (72,101 to 79,896)	1,481,906 (1,142,619 to 1,789,513)	52 (28 to 78)
Universal BEP (ce)	25,441 (14,267 to 40,379)	75,852 (72,101 to 79,896)	3,426,122 (2,642,158 to 4,030,132)	121 (70 to 196)
MMS + targeted BEP (ce)	22,541 (14,796 to 34,629)	MMS: 67,674 (64,299 to 71,418), BEP: 8,178 (7,445 to 9,033)	1,691,688 (1,351,421 to 2,002,914)	54 (32 to 77)
Universal BEP (he)	40,739 (26,320 to 58,666)	75,852 (72,101 to 79,896)	3,426,122 (2,642,158 to 4,030,132)	74 (47 to 111)
MMS + targeted BEP (he)	25,710 (17,848 to 37,635)	MMS: 67,674 (64,299 to 71,418), BEP: 8,178 (7,445 to 9,033)	1,691,688 (1,351,421 to 2,002,914)	47 (29 to 64)
Mali				
Baseline IFA	-	28,074 (25,903 to 30,983)	489,512 (387,484 to 601,238)	-
Universal MMS	14,658 (8,691 to 24,800)	75,044 (70,709 to 79,672)	1,466,118 (1,129,808 to 1,774,937)	72 (37 to 118)
Universal BEP (ce)	18,683 (11,059 to 30,763)	75,044 (70,709 to 79,672)	3,080,566 (2,568,366 to 3,720,033)	150 (84 to 249)
MMS + targeted BEP (ce)	16,608 (11,308 to 26,125)	MMS: 67,254 (63,410 to 71,465), BEP: 7,790 (7,084 to 8,658)	1,633,605 (1,355,894 to 1,945,998)	73 (40 to 104)
Universal BEP (he)	43,657 (28,017 to 61,122)	75,044 (70,709 to 79,672)	3,080,566 (2,568,366 to 3,720,033)	62 (41 to 94)
MMS + targeted BEP (he)	20,233 (13,885 to 30,779)	MMS: 67,254 (63,410 to 71,465), BEP: 7,790 (7,084 to 8,658)	1,633,605 (1,355,894 to 1,945,998)	59 (35 to 85)
Tanzania				
Baseline IFA	-	21,463 (19,789 to 23,713)	374,242 (296,461 to 460,573)	-
Universal MMS	6,010 (2,891 to 11,347)	88,464 (87,880 to 89,111)	1,728,290 (1,339,156 to 2,069,445)	253 (112 to 481)
Universal BEP (ce)	7,576 (4,079 to 12,215)	88,464 (87,880 to 89,111)	3,749,575 (3,104,093 to 4,530,882)	487 (256 to 814)
MMS + targeted BEP (ce)	6,843 (3,775 to 11,824)	MMS: 79,992 (79,045 to 81,055), BEP: 8,472 (7,728 to 9,256)	1,921,643 (1,576,711 to 2,254,872)	245 (127 to 405)
Universal BEP (he)	26,347 (17,587 to 36,812)	88,464 (87,880 to 89,111)	3,749,575 (3,104,093 to 4,530,882)	133 (88 to 195)
MMS + targeted BEP (he)	8,991 (5,664 to 14,505)	MMS: 79,992 (79,045 to 81,055), BEP: 8,472 (7,728 to 9,256)	1,921,643 (1,576,711 to 2,254,872)	181 (106 to 274)

Table 5. Population-level effect and cost-effectiveness of interventions over first 2 years of life with 95% UIs.

ICER is the difference in cost between 2 interventions, divided by the difference in their benefit.

BEP, balanced energy protein; ce, scenarios with current evidence effects for BEP; DALY, disability-adjusted life year; he, scenarios with hypothesized effects for BEP; ICER, incremental cost-effectiveness ratio; IFA, iron and folic acid; MMS, multiple micronutrient supplementation; UI, uncertainty interval.

https://doi.org/10.1371/journal.pmed.1003902.t005



Fig 5. Incremental cost per DALY averted among the first 2 years of life in each modeled location for each simulated scenario relative to the baseline scenario (log-scale). BEP, balanced energy protein; ce, scenarios with current evidence effects for BEP; DALY, disability-adjusted life year; he, scenarios with hypothesized effects for BEP; MMS, multiple micronutrient supplementation.

https://doi.org/10.1371/journal.pmed.1003902.g005

effective in the sensitivity analysis, with \$56 (95% UI: 38, 81), \$29 (95% UI: 18, 49), \$35 (95% UI: 22, 62), and \$135 (95% UI: 112, 415) per DALY averted relative to baseline in the respective locations in the MMS + targeted BEP scenario using ce effect sizes.

Discussion

Inadequate maternal nutritional intake is a major contributor to poor birth outcomes [7]. To improve birth outcomes, WHO recommends antenatal supplementation with multiple micronutrients that include IFA in the context of rigorous research [8] and BEP in populations with over 20% burden of undernourished pregnant women [14]. As countries consider investing in MMS, there are opportunities to integrate supplement delivery with optimized formulations of BEP containing MMS, so that micronutrient and energy deficits can be met within a single

Table 6.	GDP per capita and cost-effectiveness thresholds in 2020 USD with simulated cost-effectiveness ratios in 2021 USD per DALY averted relative to	baseline
with 95%	UIs.	

Country	GDP per capita [37]	Half of GDP per capita	Incremental cost of universal MMS per DALY averted relative to baseline	Incremental cost of MMS + targeted BEP (ce) per DALY averted relative to baseline	Incremental cost of MMS + targeted BEP (he) per DALY averted relative to baseline
India	\$2,108	\$1,054	70 (43 to 104)	83 (58 to 111)	67 (47 to 88)
Pakistan	\$1,285	\$643	52 (28 to 78)	54 (32 to 77)	47 (29 to 64)
Mali	\$891	\$446	72 (37 to 118)	73 (40 to 104)	59 (35 to 85)
Tanzania	\$1,122	\$561	253 (112 to 481)	245 (127 to 405)	181 (106 to 274)

BEP, balanced energy protein; ce, current evidence effects for BEP; DALY, disability-adjusted life year; GDP, gross domestic product; he, hypothesized effects for BEP; UI, uncertainty interval.

https://doi.org/10.1371/journal.pmed.1003902.t006



Fig 6. Incremental cost and DALYs averted among the first 2 years of life per 100,000 births for each simulated scenario relative to the baseline scenario in each modeled location. *Each point in this figure represents one of the 100 simulation runs performed, each of which used varying parameter values within the parameter UIs used in our model. BEP, balanced energy protein; ce, scenarios with current evidence effects for BEP; DALY, disability-adjusted life year; he, scenarios with hypothesized effects for BEP; MMS, multiple micronutrient supplementation; UI, uncertainty interval.

https://doi.org/10.1371/journal.pmed.1003902.g006

vehicle. While some cost-effectiveness analyses have shown that switching from IFA to MMS is very cost-effective in some countries [10-13], there are no detailed studies, to the best of our knowledge, estimating the cost-effectiveness of BEP or comparing the cost-effectiveness of MMS and BEP [26]. Our study modeled the currently available evidence from Cochrane Systematic Reviews on antenatal IFA, MMS, and BEP supplementation as well as hypothesized effects of BEP as suggested by some evidence in a best-case scenario. We explored the impact of targeting undernourished women for BEP and found that MMS + targeted BEP averts more DALYs and remains cost-effective compared to universal MMS. Targeting may be an attractive and cost-effective strategy to consider, especially in countries that do not meet WHO's 20% undernourishment prevalence. The strength of our approach came from using dynamic, individual-based microsimulation that leveraged draw-level estimates from the 2017 GBD study to compare the impact on DALYs and cost-effectiveness of different supplementation scenarios. Our study provides decision-makers a side-by-side comparison of the incremental cost-effectiveness of scaling strategies for MMS and BEP compared to current baseline levels of antenatal IFA consumption in 4 countries with a considerable burden of undernourishment among pregnant women.

Our ICERs for universal MMS were greater than a recent cost-effectiveness analysis of antenatal MMS supplementation, which estimated \$22.47, \$22.64, and \$61.82 per DALY for India, Pakistan, and Tanzania, respectively (no data available for Mali) using the MMS Cost Benefit Tool [10]. However, the MMS Cost Benefit Tool only considered commodity costs which may have underestimated the costs. Both studies showed an approximate 3 times greater cost per DALY in Tanzania than in India and Pakistan. These results may reflect the differences in baseline characteristics of the populations: Since nutritional supplementations mainly affect birth weight, the greatest gains were made in countries with higher baseline prevalence of LBW and higher percentage of DALYs attributable to LBW. Tanzania has an overall higher mean birth weight at baseline, which means additional improvement in birth weight from supplementation may not have as great an impact on averting DALYs.

Targeting BEP for low BMI mothers and providing MMS to non-low BMI mothers averts more DALYs and is as cost-effective as universal MMS when using the currently available evidence for BEP effects. When considering the hypothesized best-case effects of BEP, MMS + targeted BEP averts more DALYs and is even more cost-effective than universal MMS. WHO has proposed using 1 to 3× GDP per capita as the cost-effective threshold in recommending adoption of new interventions; however, some have argued that this WHO threshold may still be too expensive and thresholds of half of GDP per capita are more reasonable [37]. Average GDP per capita (2020 USD) in 2019 for low-income countries was \$780 while that for lowermiddle income countries was \$2180 [38]. Under these thresholds, targeted BEP scenarios were very cost-effective at under \$100 per DALY for India, Pakistan, and Mali but over \$100 threshold at \$245 per DALY for Tanzania.

ICERs for universal BEP relative to baseline using the ce for BEP effects were consistently higher than ICERs for universal MMS relative to baseline. Horton and colleagues ranked the cost-effectiveness of 93 interventions for low- and middle-income countries and placed BEP as potentially appropriate for consideration in upper-middle countries at \$500 per DALY [22]. Among modeled countries using ce effects, our estimates for BEP were under \$500 per DALY, although the upper bound of the UI was greater than \$500 for Tanzania. When considering the hypothesized effects of BEP on birth weight and its effects on child wasting and stunting in the best-case scenario, the ICERs for universal BEP relative to baseline were lower than the ICERs for universal MMS in Mali and Tanzania, but remained higher in India and Pakistan. This trend of greater relative cost-effectiveness of universal BEP under the hypothesized effects of BEP is consistent with the greater proportion of combined burden attributable to child stunting and wasting in our modeled locations. This suggests that if the hypothesized effects of BEP on child wasting and stunting were realized, it may be more cost-effective in locations with high wasting and stunting burdens. Still, increasing caloric intake through BEP for all women is expensive. Additionally, the prevalence of obesity and overweight is increasing in low- and middle-income countries; among our modeled countries, the prevalence of overweight and obese women ranged from 20% to 50% (S2 Supplement). Women with overweight or obesity are at higher risk of fetal overgrowth and macrosomia (\geq 4,000 g) [23], and the effect BEP supplementation among this group is still unknown.

Our study is limited by some key assumptions. First, the effect size of BEP on birth weight may be hard to generalize across different settings due to heterogeneity in the BEP product supplementation period and comparison groups used in the trials. Studies assessing the effect of BEP have used different supplement compositions such as chocolate-flavored liquid, biscuits, dry powder, and combination of foods such as milk, bread, oil, beans, and maize [15]. This makes it difficult to put a standard formula or price on BEP if standard formulas are not used. Cost and acceptability of the BEP supplements will depend on local preferences and availability. Different studies also had different supplementation durations, which may affect

both intervention cost and impact. Additionally, studies varied in the comparison interventions in the control groups: some studies used alternative interventions such as vitamins and minerals only or low-energy protein source, placebo pills, or no interventions as the control group. Nonetheless, we assumed that the effect size reported in the literature is that of BEP compared to MMS.

Second, we applied our interventions to 90% of women who attend ANC, which may not be easy to achieve without substantial effort. The proportion of women who took antenatal iron for 90 or more days is low (20% to 40%) among our modeled countries. Multiple barriers that affect the effective coverage of IFA include pregnant women's beliefs that hinder early ANC attendance, poor healthcare worker performance, verticalization of healthcare delivery, lack of availability of commodities, and irregular supply of essential pharmaceuticals [39]. Our sensitivity analysis estimated a lower population-level impact if intervention coverages only achieved the level of baseline IFA coverage. Additionally, timing of ANC attendance also varies and those who attend late in pregnancy would not receive maximum benefit. Extra costs may be needed to encourage women to attend ANC early to achieve effect sizes seen in the trials. Adherence to daily supplementation may also be lower in real-world conditions than observed in trials (although palatable and MMS-integrated BEP formulations might improve adherence). Additionally, our model does not consider differential utilization of ANC services by factors such as residency (urban versus rural) or maternal age, although such differences have been documented [40]. If supplementation interventions do not reach higher risk groups such as rural and/or young mothers, our model may overestimate their impact. Interventions will need to address local health system constraints and conduct robust implementation research to improve the uptake and adherence to supplementation programs.

Third, current available evidence may have underestimated the effect of BEP. The small improvement in birth weight associated with BEP may reflect the small net increment in energy intake achieved in the trials [15]. Substitution of the BEP supplement for the main diet, as well as nonadherence, has been postulated as possible explanations [15]. In the Gambia, high-energy groundnut biscuits provided to chronically undernourished women antenatally improved birth weight by much more: 136 g compared with those not supplemented antenatally [16]. The authors argued that supplementation is most effective when there is a large energy gap to fill [16,41]. While we attempted to address the greater magnitude of BEP's impact among babies at the lower end of the birth weight distribution by stratifying birth weight by maternal undernourishment status, we may not have captured the full extent of the additional benefit that BEP may have for nutritionally vulnerable pregnancies. Further, the effect of maternal supplementation regimens has been shown to be greater among anemic mothers [42], which we did not consider in our simulation. Therefore, our modeled interventions may be more effective in countries with high maternal anemia burdens, which is not reflected in our results. Additionally, BEP's impact on long-term physical growth has not been thoroughly examined [15,20]. One study from Indonesia found greater weight and length growth in children born to mothers who were given high-energy supplements compared to those born to mothers given a low-energy supplement [19]. More high-quality studies with long-term follow-up are needed to draw conclusions for the effect of BEP on birth weight and child growth. Finally, reviews of MMS and BEP do not show strong evidence for reduction in preterm births or an increase in average gestational age (for BEP) [9,15]. Hence, we assume the improvement in LBW (<2,500 g) reported in the literature is captured by improved birth weight independent of gestational age and we did not model any effect on gestational weeks. Notably, while we considered uncertainty in the parameters used in this model, which are reflected in the UIs of our results, we did not attempt to quantify structural uncertainty regarding the relationships between parameters nor heterogeneity at the subnational level in our model.

Fourth, our cost model accounts only for the supplementation period of 6 months for IFA, MMS, and BEP; we did not account for varying duration of supplementation, which may depend on timing of ANC visits if supplements were distributed through facilities. We only considered the costs to payers and did not consider other societal costs, such as time spent attending ANC for women. However, since ANC has multiple benefits, these costs may be shared among other facility-based antenatal interventions.

Finally, our model only estimated the effect of supplementation on outcomes among children under 2 and did not consider potential effects on maternal outcomes or outcomes among children that occur beyond the first 2 years of life. If we had included maternal outcomes as well, the interventions are likely to be more cost-effective.

Our study shows that MMS in combination with targeted BEP averts more DALYs than universal MMS compared to baseline IFA while remaining cost-effective (under \$100 per DALY for India, Pakistan, and Mali and under \$250 per DALY for Tanzania). As countries begin to consider using MMS in alignment with recently updated WHO guidelines, targeted BEP could be considered as an additional cost-effective strategy to maximize benefit and synergize program implementation.

Supporting information

S1 Supplement. Model development workflow. (DOCX)

S2 Supplement. Meta-analysis of maternal BMI and infant birth weight. BMI, body mass index.

(DOCX)

S3 Supplement. Meta-analysis of birth weight shift of multiple micronutrients versus iron with or without folic acid control. (DOCX)

S4 Supplement. CHEERS checklist. CHEERS, Consolidated Health Economic and Evaluation Reporting Standards.

(DOCX)

S5 Supplement. Model verification plots comparing GBD study inputs and simulation outputs. GBD, Global Burden of Disease. (DOCX)

S6 Supplement. Sensitivity analysis figures. (DOCX)

Acknowledgments

We would like to thank Dr. Laura Lamberti (Deputy Director) and Dr. Rebecca Brander (Program Officer) at the Bill and Melinda Gates Foundation for their contribution in the conceptualization and development of this project as well as their review of the manuscript.

Author Contributions

Conceptualization: Nicole Young, Alison Bowman, James Collins, Abraham D. Flaxman.

Data curation: Nicole Young, Christopher Troeger.

Formal analysis: Nicole Young, Alison Bowman, Christopher Troeger, Abraham D. Flaxman.

Investigation: Nicole Young, Alison Bowman, James Collins, Abraham D. Flaxman.

Methodology: Nicole Young, Alison Bowman, James Collins, Nathaniel D. Blair-Stahn, Christopher Troeger, Abraham D. Flaxman.

Project administration: James Collins, Abraham D. Flaxman.

Software: Kjell Swedin, James Collins, Abraham D. Flaxman.

Supervision: James Collins, Paulina A. Lindstedt, Abraham D. Flaxman.

Validation: Nicole Young, Alison Bowman, Kjell Swedin, James Collins, Abraham D. Flaxman.

Visualization: Nicole Young, Abraham D. Flaxman.

Writing - original draft: Nicole Young.

Writing – review & editing: Nicole Young, Alison Bowman, Kjell Swedin, James Collins, Paulina A. Lindstedt, Christopher Troeger, Abraham D. Flaxman.

References

- Kind KL, Moore VM, Davies MJ. Diet around conception and during pregnancy—effects on fetal and neonatal outcomes. Reprod Biomed Online. 2006 May; 12(5):532–41. https://doi.org/10.1016/s1472-6483(10)61178-9 PMID: 16790095
- Mandruzzato G, Antsaklis A, Botet F, Chervenak FA, Figueras F, Grunebaum A, et al. Intrauterine restriction (IUGR). J Perinat Med. 2008; 36(4):277–81. <u>https://doi.org/10.1515/JPM.2008.050</u> PMID: 18598115
- Institute of Medicine (US) Committee on Improving Birth Outcomes. Improving Birth Outcomes: Meeting the Challenge in the Developing World [Internet]. Bale JR, Stoll BJ, Lucas AO, editors. Washington (DC): National Academies Press (US); 2003 [cited 2021 Feb 8]. Available from: <u>http://www.ncbi.nlm.nih.gov/books/NBK222097/</u>.
- Victora CG, Adair L, Fall C, Hallal PC, Martorell R, Richter L, et al. Maternal and child undernutrition: consequences for adult health and human capital. Lancet. 2008 Jan 26; 371(9609):340–57. https://doi. org/10.1016/S0140-6736(07)61692-4 PMID: 18206223
- Blencowe H, Krasevec J, de Onis M, Black RE, An X, Stevens GA, et al. National, regional, and worldwide estimates of low birthweight in 2015, with trends from 2000: a systematic analysis. Lancet Glob Health. 2019 May 15; 7(7):e849–60. https://doi.org/10.1016/S2214-109X(18)30565-5 PMID: 31103470
- 6. Ritchie H, Roser M. Micronutrient Deficiency. Our World Data [Internet]. 2017 Aug 11 [cited 2020 May 13]; Available from: https://ourworldindata.org/micronutrient-deficiency.
- Black RE, Victora CG, Walker SP, Bhutta ZA, Christian P, de Onis M, et al. Maternal and child undernutrition and overweight in low-income and middle-income countries. Lancet. 2013 Aug 3; 382 (9890):427–51. https://doi.org/10.1016/S0140-6736(13)60937-X PMID: 23746772
- Tuncalp Ö, Rogers LM, Lawrie TA, Barreix M, Peña-Rosas JP, Bucagu M, et al. WHO recommendations on antenatal nutrition: an update on multiple micronutrient supplements. BMJ Glob Health. 2020 Jul; 5(7):e003375. https://doi.org/10.1136/bmjgh-2020-003375 PMID: 32732247
- Keats EC, Haider BA, Tam E, Bhutta ZA. Multiple-micronutrient supplementation for women during pregnancy. Cochrane Database Syst Rev. 2019 14; 3:CD004905. https://doi.org/10.1002/14651858. CD004905.pub6 PMID: 30873598
- MMS cost benefit tool: integration of who meta-analyses draft technical report. [Internet]. 2019 [cited 2021 Feb 15]. Available from: https://www.nutritionintl.org/learning-resources-home/mms-cost-benefittool/.
- Svefors P, Selling KE, Shaheen R, Khan AI, Persson L-Å, Lindholm L. Cost-effectiveness of prenatal food and micronutrient interventions on under-five mortality and stunting: Analysis of data from the MINIMat randomized trial, Bangladesh. PLoS ONE. 2018; 13(2):e0191260. https://doi.org/10.1371/ journal.pone.0191260 PMID: 29447176
- Engle-Stone R, Kumordzie SM, Meinzen-Dick L, Vosti SA. Replacing iron-folic acid with multiple micronutrient supplements among pregnant women in Bangladesh and Burkina Faso: costs, impacts, and cost-effectiveness. Ann N Y Acad Sci. 2019 May; 1444(1):35–51. https://doi.org/10.1111/nyas.14132 PMID: 31134641

- Kashi B, M Godin C, Kurzawa ZA, Verney AMJ, Busch-Hallen JF, De-Regil LM. Multiple Micronutrient Supplements Are More Cost-effective Than Iron and Folic Acid: Modeling Results from 3 High-Burden Asian Countries. J Nutr. 2019 Jul 1; 149(7):1222–9. https://doi.org/10.1093/jn/nxz052 PMID: 31131412
- 14. WHO recommendations on antenatal care for a positive pregnancy experience. Geneva: World Health Organization; 2016.
- Ota E, Hori H, Mori R, Tobe-Gai R, Farrar D. Antenatal dietary education and supplementation to increase energy and protein intake. Cochrane Database Syst Rev [Internet]. 2015 [cited 2020 Apr 15]; (6). Available from: http://www.cochranelibrary.com/cdsr/doi/10.1002/14651858.CD000032.pub3/full. PMID: 26031211
- Ceesay SM, Prentice AM, Cole TJ, Foord F, Weaver LT, Poskitt EM, et al. Effects on birth weight and perinatal mortality of maternal dietary supplements in rural Gambia: 5 year randomised controlled trial. BMJ. 1997 Sep 27; 315(7111):786–90. https://doi.org/10.1136/bmj.315.7111.786 PMID: 9345173
- Shaheen R, Shaheen R, de Francisco A, El Arifeen S, Ekström E-C, Persson LÅ. Effect of prenatal food supplementation on birth weight: an observational study from Bangladesh. Am J Clin Nutr. 2006 Jun 1; 83(6):1355–61. https://doi.org/10.1093/ajcn/83.6.1355 PMID: 16762947
- Lassi ZS, Padhani ZA, Rabbani A, Rind F, Salam RA, Das JK, et al. Impact of Dietary Interventions during Pregnancy on Maternal, Neonatal, and Child Outcomes in Low- and Middle-Income Countries. Nutrients. 2020 Feb; 12(2):531.
- Kusin JA, Kardjati S, Houtkooper JM, Renqvist UH. Energy supplementation during pregnancy and postnatal growth. Lancet. 1992 Sep 12; 340(8820):623–6. https://doi.org/10.1016/0140-6736(92) 92168-f PMID: 1355209
- Stevens B, Buettner P, Watt K, Clough A, Brimblecombe J, Judd J. The effect of balanced protein energy supplementation in undernourished pregnant women and child physical growth in low- and middle-income countries: a systematic review and meta-analysis: Undernourished pregnant women and child growth. Matern Child Nutr. 2015 Oct; 11(4):415–32. https://doi.org/10.1111/mcn.12183 PMID: 25857334
- World Food Programme. Fill the Nutrient Gap [Internet]. 2020 [cited 2021 Feb 8]. Available from: https:// www.wfp.org/publications/2020-fill-nutrient-gap.
- Horton S, Gelband H, Jamison D, Levin C, Nugent R, Watkins D. Ranking 93 health interventions for low- and middle-income countries by cost-effectiveness. PLoS ONE. 2017 Aug 10; 12(8):e0182951. https://doi.org/10.1371/journal.pone.0182951 PMID: 28797115
- Liu L, Ma Y, Wang N, Lin W, Liu Y, Wen D. Maternal body mass index and risk of neonatal adverse outcomes in China: a systematic review and meta-analysis. BMC Pregnancy Childbirth. 2019 Mar 29; 19 (1):105. https://doi.org/10.1186/s12884-019-2249-z PMID: 30922244
- Bhutta ZA, Das JK, Rizvi A, Gaffey MF, Walker N, Horton S, et al. Evidence-based interventions for improvement of maternal and child nutrition: what can be done and at what cost? Lancet. 2013 Aug 3; 382(9890):452–77. https://doi.org/10.1016/S0140-6736(13)60996-4 PMID: 23746776
- Keats EC, Das JK, Salam RA, Lassi ZS, Imdad A, Black RE, et al. Effective interventions to address maternal and child malnutrition: an update of the evidence. Lancet Child Adolesc Health [Internet]. 2021 Mar 7 [cited 2021 Mar 15]; Available from: <u>https://www.sciencedirect.com/science/article/pii/</u> S2352464220302741.
- WHO. Balanced energy and protein supplementation during pregnancy [Internet]. 2019 [cited 2020 Sep 17]. Available from: https://www.who.int/elena/titles/energy_protein_pregnancy/en/
- 27. Simulation Science Team. Vivarium Documentation [Internet]. 2017 [cited 2020 Oct 26]. Available from: https://vivarium.readthedocs.io/en/latest/.
- Stanaway JD, Afshin A, Gakidou E, Lim SS, Abate D, Abate KH, et al. Global, regional, and national comparative risk assessment of 84 behavioural, environmental and occupational, and metabolic risks or clusters of risks for 195 countries and territories, 1990–2017: a systematic analysis for the Global Burden of Disease Study 2017. Lancet. 2018 Nov 10; 392(10159):1923–94. https://doi.org/10.1016/ S0140-6736(18)32225-6 PMID: 30496105
- 29. Murray CJL, Callender CSKH, Kulikoff XR, Srinivasan V, Abate D, Abate KH, et al. Population and fertility by age and sex for 195 countries and territories, 1950–2017: a systematic analysis for the Global Burden of Disease Study 2017. Lancet. 2018 Nov 10; 392(10159):1995–2051. https://doi.org/10.1016/S0140-6736(18)32278-5 PMID: 30496106
- 30. James SL, Abate D, Abate KH, Abay SM, Abbafati C, Abbasi N, et al. Global, regional, and national incidence, prevalence, and years lived with disability for 354 diseases and injuries for 195 countries and territories, 1990–2017: a systematic analysis for the Global Burden of Disease Study 2017. Lancet. 2018 Nov 10; 392(10159):1789–858. https://doi.org/10.1016/S0140-6736(18)32279-7 PMID: 30496104
- Roth GA, Abate D, Abate KH, Abay SM, Abbafati C, Abbasi N, et al. Global, regional, and national agesex-specific mortality for 282 causes of death in 195 countries and territories, 1980–2017: a systematic

analysis for the Global Burden of Disease Study 2017. Lancet. 2018 Nov 10; 392(10159):1736–88. https://doi.org/10.1016/S0140-6736(18)32203-7 PMID: 30496103

- Peña-Rosas JP, De-Regil LM, Garcia-Casal MN, Dowswell T. Daily oral iron supplementation during pregnancy. Cochrane Database Syst Rev. 2015 Jul 22; 7:CD004736.
- 33. MAL-ED Network Investigators. The MAL-ED study: a multinational and multidisciplinary approach to understand the relationship between enteric pathogens, malnutrition, gut physiology, physical growth, cognitive development, and immune responses in infants and children up to 2 years of age in resourcepoor environments. Clin Infect Dis. 2014; 1(59 Suppl 4):S193–206. https://doi.org/10.1093/cid/ciu653 PMID: 25305287
- Scott N, Delport D, Hainsworth S, Pearson R, Morgan C, Huang S, et al. Ending malnutrition in all its forms requires scaling up proven nutrition interventions and much more: a 129-country analysis. BMC Med. 2020 Nov 13; 18(1):356. https://doi.org/10.1186/s12916-020-01786-5 PMID: 33183301
- **35.** Shekar M, Kakietek J, Eberwein JD, Walters D. An Investment Framework for Nutrition: Reaching the Global Targets for Stunting, Anemia, Breastfeeding, and Wasting. World Bank Publications. 2017:241 p.
- **36.** US Bureau of Labor Statistics. CPI Inflation Calculator [Internet]. [cited 2021 Sep 20]. Available from: https://www.bls.gov/data/inflation_calculator.htm.
- Ochalek JM, Lomas J, Claxton KP. Cost per DALY averted thresholds for low- and middle-income countries : evidence from cross country data. CHE Research Paper, nos. 122, Centre for Health Economics, University of York, York, UK, pp. 1–50:51.
- The World Bank. GDP per capita (current US\$) [Internet]. 2019 [cited 2020 Oct 1]. Available from: https://data.worldbank.org/indicator/NY.GDP.PCAP.CD.
- Siekmans K, Roche M, Kung'u JK, Desrochers RE, De-Regil LM. Barriers and enablers for iron folic acid (IFA) supplementation in pregnant women. Matern Child Nutr. 2018; 14(S5):e12532. https://doi. org/10.1111/mcn.12532 PMID: 29271115
- Langa N, Bhatta T. The rural-urban divide in Tanzania: Residential context and socioeconomic inequalities in maternal health care utilization. PLoS ONE. 2020; 15(11):e0241746. <u>https://doi.org/10.1371/</u> journal.pone.0241746 PMID: 33166310
- Prentice AM, Cole TJ, Foord FA, Lamb WH, Whitehead RG. Increased birthweight after prenatal dietary supplementation of rural African women. Am J Clin Nutr. 1987 Dec; 46(6):912–25. https://doi.org/10. 1093/ajcn/46.6.912 PMID: 3687824
- 42. Smith ER, Shankar AH, Wu LS-F, Aboud S, Adu-Afarwuah S, Ali H, et al. Modifiers of the effect of maternal multiple micronutrient supplementation on stillbirth, birth outcomes, and infant mortality: a meta-analysis of individual patient data from 17 randomised trials in low-income and middle-income countries. Lancet Glob Health. 2017 Nov 1; 5(11):e1090–100. https://doi.org/10.1016/S2214-109X(17) 30371-6 PMID: 29025632